Author's response to reviews

Title: Changes in mental health services and suicide mortality in Norway: an ecological study

Authors:

Haakon A. Johannessen (hajn@fhi.no)
Gudrun Dieserud (gudrun.dieserud@fhi.no)
Bjorgulf Claussen (bjorgulf.claussen@medisin.uio.no)
Per-Henrik Zahl (per-henrik.zahl@fhi.no)

Version: 2 Date: 12 March 2010

Author's response to reviews: see over
Dear Editor, BMC Public Health

Please consider the enclosed manuscript “Changes in mental health services and suicide mortality in Norway: an ecological study”. The paper has not previously been published, either in whole or in part. No related paper is in press or under review elsewhere.

The following text has been added to the method section of the changed manuscript:

All data used in the present study are openly available and can be downloaded at Statistics Norway and Norwegian Institute for Alcohol and Drug Research, except data on sales of antidepressants which can be provided by the wholesales register in Norway by request.

No other changes have been made.

Sincerely,

Håkon A. Johannessen